Author's response to reviews

Title: Protocol for a double-blind randomised placebo-controlled trial of Lithium Carbonate in patients with Amyotrophic Lateral Sclerosis (LiCALS) [EudraCT number: 2008-006891-31]

Authors:

Ammar Al-Chalabi (ammar.al-chalabi@kcl.ac.uk)
Pamela J Shaw (pamela.shaw@sheffield.ac.uk)
Carolyn A Young (carolyn.young@thewaltoncentre.nhs.uk)
Karen E Morrison (k.morrison@bham.ac.uk)
Caroline Murphy (caroline.murphy@kcl.ac.uk)
Marie Thornhill (marie.thornhill@kcl.ac.uk)
Joanna Kelly (joanna.kelly@kcl.ac.uk)
I Nick Steen (nick.steen@newcastle.ac.uk)
P Nigel Leigh (n.leigh@kcl.ac.uk)
on behalf of UKMND-LiCALS (ammar.al-chalabi@kcl.ac.uk)

Version: 5 Date: 19 August 2011

Author's response to reviews:

Two typos spotted - LiCO3 - the 3 should be subscript, and inconsistency in spelling randomisation or its derivatives - sometimes with z, sometimes with s. Now made consistently s.